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• We are trying to fix a problem with the muon/hadron monitors data quality
» Data has many repetitions 
» Sometimes pedestal gets the signal values
» This issue is very high in rep rate 1.2 sec operations
» Thanking to Phil, Katsuya, Dan, Beau, Max, Jason and Jianming for helping 

on solving this issue!
• Dan and Jeremy replaced the CAMAC 377 timer card for muon monitors
• There was a work completed on the temperature control valve and chilled water 

flow on the Horn1 RAW system (Ref: 238060): Thanks to the team!
» The team found and fixed several issues and adjusted the chilled water 

flow.
» This work helps to solve the high temperature issue.
» We keep monitoring this issue with high beam power operations.

• Studies on baffle temperature issue are on going

TSD report for AD 9 am ops meeting

https://www-bd.fnal.gov/Elog/?orEntryId=238060

